Author’s response to reviews

Title: TTFields alone and in combination with chemotherapeutic agents effectively reduce the viability of MDR cell sub-lines that over-express ABC transporters

Authors:

Rosa S Schneiderman (roza@novo-cure.com)
Esther Shmueli (shmuelieti@hotmail.com)
Eilon D Kirson (eilon@novo-cure.com)
Yoram Palti (Yoram@novo-cure.com)

Version: 3 Date: 17 May 2010

Author’s response to reviews: see over
TTFields alone and in combination with chemotherapeutic agents effectively reduce the viability of MDR cell sub-lines that over-express ABC transporters.

Rosa S Schneiderman, Esther Shmueli, Eilon D Kirson and Yoram Palti

**Formatting changes made**

Major revisions

1. Abstract:
   a. Page 2, line 5: the sentence was re-phrased;
   b. Page 2, line 17: added a sentence relating to statistical analysis of the results.

2. Authors’ contribution: final manuscript was approved by all authors.

Minor revisions

1. Tables:
   a. The table titles are one sentence long.
   b. The legends are now placed below each table.
   c. Vertical lines were removed from tables.

2. Figures:
   a. Figures were cropped.
   b. Figure names/numbers were removed from the image files.

3. Typography: the manuscript was checked for typographical errors.